Massachusetts Institute of Technology Handout 4 6.046J/18.410J: Introduction to Algorithms February 7, 2002 Professors Michel Goemans and Piotr Indyk

Readings on Program Verification

• First chapter of Jeffrey Kingston’s book Algorithms and Data Structures: Design, Correctness, Analysis. Addison-Wesley Publishing Co., 1991. This is required reading.

Due to copyright concerns we will not make this chapter available via the web. If you missed the in-class handout, you can obtain a copy from the course secretary.

Bring this handout to recitation on Friday, February 8.